CS 350 Theory of Computation Assignment 2 (8 marks) Question 1 (2 marks)

This part of the assignment is on Rocky’s hypertext book (http://www.cs.montana.edu/ross/classes/spring-2006/cs350/Assignments/assignment01.htm), you must finish this question using the corresponding Applet and follow the directions there. (Click ’here’ on that page if you need any help.) This part of the assignment is due Friday at 10pm, Feb 3, 2006. Question 2 (1 marks)

Problem 1.12.b (page 85—all the questions are referred to the 1st edition of the textbook).

Question 3 (2 marks)

Problem 1.14.a, 1.14.b (page 86). Question 4 (1 marks)

Problem 1.16.b (page 86). Question 5 (2 marks)

Problem 1.23.a, 1.23.c (page 88).

Date Due: before the end of class on Wednesday, Feb 22, 2006. Late assignment will lose 2 marks for each overdue day.